Author's response to reviews

Title: BRCA1 and ERCC1 mRNA levels are associated with lymph node metastasis in Chinese patients with colorectal cancer

Authors:

yuanming LU (yuanming_lu@hotmail.com)
Baorong Song (sbr@163.com)
Jingyan Gao (gaojy@163.com)
Junjie Peng (pengjj@163.com)
Ye Xu (Xuye@163.com)
Sanjun Cai (bobojpn@hotmail.com)

Version: 6 Date: 4 December 2012

Author's response to reviews: see over
Dear Editor:

This manuscript (Manuscript ID 1187978867373802) has been revised by a native-English speaker with scientific expertise, Edanz, I send here the Edit_TrackedCopy from the Edanz and 1187978867373802 revised manuscript (ClearCopy) to you, please check and keep in touch, thank you.

Sincerely,

Cai Sanjun

Dec. 4th, 2012